Author's response to reviews

Title: Acute Retroperitoneal Bleeding due to Inferior Mesenteric Artery Aneurysm

Authors:

Pablo Pérez-Vallecillos (vallecillos@hotmail.com)
Raquel Conde-Muiño (rcondemuinho@gmail.com)
Pablo Palma (pablopalma@andaluciajunta.es)
Inmaculada Segura-Jimenez (inma_aguasmansas@yahoo.es)
Nicolás Maldonado-Fernandez (nicovasc@hotmail.com)
Vicente García-Róspide (vgrospide@seacv.org)
José Antonio Ferrón (josea.ferron.sspa@juntadeandalucia.es)

Version: 2 Date: 21 January 2010

Author's response to reviews: see over
Dear Editor:

Enclosed please find our case report:

“Acute retroperitoneal bleeding due to inferior mesenteric artery aneurism”

Manuscript corrections have been made upon reviewer’s report as follows:

1. The surgical technique was better and further explained as suggested by Theodore H Teruya. Furthermore, the size of the CT hematoma was included.
2. The background was reorganized as indicated by Raffaele Pulli.
3. Further emphasis was also placed on endovascular techniques, this being discussed thereafter (suggested by Onur Goksel)
4. The paper has been thoroughly revised concerning its English

We report on an unexpected and rare association between abdominal pain and visceral aneurysm. The case was solved by surgery with the involvement of vascular and colorectal specialists of our University Hospital. The description is accompanied with a review of the literature and excellent CT-imaging and should be of interest for gastroenterologists and surgeons.

We will be very honored to see it published in BMC gastroenterology

Please don’t hesitate to contact us for further corrections or comments.

Yours sincerely

Pablo Palma, M.D., E.B.S.Q.C.
Head Division of Colon & Rectal Surgery
HUVN Granada
Spain
pablopalma@andaluciajunta.es